Craig Chambers CSE 505 (Fall ’95) Homework Assignments

Homework Assignment #5 Due Friday 11/3

From Chapter 5 exercises:

3.1: don’t assign type attributes, other than the formals of parameterized abstractions; treat modules as lazily evaluated 5.1: for lazy evaluation only 5.2 5.7abc: don’t worry about equivalence parts 8.1: use this syntax for cases: cases E_0 of (I_1:θ_1) E_1; (I_2:θ_2) E_2; don’t worry with copy rules; also explain why the extended language no longer has unicity of typing.

From Chapter 6 exercises: 1.3ab 1.6ab